Incentivizing Research & Innovation with Agrobiodiversity Conserved In Situ: Possibilities and Limitations of a Blockchain-Based Solution

Mrinalini Kochupillai* Ulrich Gallersdörfer Julia Köninger Roman Beck Department of Aerospace and Geodesy Department of Biology Department of BusinessIT Department of Informatics Technical University of Munich Technical University of Munich Universidad de Vigo IT University of Copenhagen Copenhagen, Denmark Munich, Germany Munich, Germany Vigo, Spain ulrich.gallersdoerfer@tum.de julia.koninger@uvigo.es beck@itu.dk m.kochupillai@tum.de

Abstract-This article explores the use of blockchain for agrobiodiversity (B4A) with a specific focus on (i) providing an overview of the existing regulatory challenges when it comes to conserving agrobiodiversity, which results in a lack of research and innovation when it comes to agrobiodiversity conserved in situ, (ii) investigating how a blockchain-based solution may help overcome these challenges, and (iii) illustrating how incentive mechanisms can help to overcome existing intellectual property regimes that prevent effective conservation, research and innovation (CRI). Our research identifies (i) lack of incentives, (ii) lack of trust among stakeholders, and (iii) lack of traceability options as main hindering reasons for in situ CRI with agrobiodiversity. Further, We find that blockchain solutions may empower data providers, including small farmers, to collectively track, control and monetize the use of data and assets shared, while minimizing fraudulent activities. Transaction costs may also be lowered by removing complex and expensive interaction processes. However, further research and development is necessary to design an ethical and sustainable blockchain-based solution to incentivize in situ conservation, research and innovation with agrobiodiversity. Some future directions of research are recommended.

Index Terms—Agrobiodiversity, Blockchain, Incentives, Sustainability, Benefits

I. INTRODUCTION

Biodiversity is defined as "the variability among living organisms from all sources ... and the ecological complexes of which they are part; this includes diversity within species, between species and of ecosystems" (CBD). The term biodiversity includes terrestrial and aquatic biodiversity, as well as agricultual biodiversity (agrobiodiversity) [1]. This article focuses on agrobiodiversity.

Agrobiodiversity comprises the biological diversity within the soil (the soil microbiome) as well as on the soil (genetic diversity contained in indigenous or local, non-uniform seeds, pollinators etc.). Efforts to conserve agrobiodiversity in its natural surroundings, such as through farming activities, is called in situ conservation.

Farming systems that utilize in and on soil agrobiodiversity are often more robust in the face of local biotic and abiotic stresses [2]. Agrobiodiversity, therefore, actively contributes to food and nutritional security in the face of rapid climate change [3]–[5].

International instruments, such as the Convention on Biological Diversity (CBD) [6] and the International Treaty on Plant Genetic Resources for Food and Agriculture ("the Seed Treaty") [7], emphasize the relevance of in situ agrobiodiversity conservation. They also require member states to adopt so called "Access and Benefit Sharing (ABS)" regimes. ABS regimes aim to facilitate research and innovation with agrobiodiversity on one hand, and grant monetary benefits (so called "benefit sharing"), to those granting access to their bio resources. However, due to several regulatory issues, the ABS system has failed to incentivize and promote in situ conservation, research and innovation (CRI) with agrobiodiversity [8, 9].

Since the 1900s, an estimated 75 per cent of crop genetic diversity has been lost [10] not at least due to expanding farm areas under commercially available uniform seeds. Additionally, as a consequence of the intense use of inorganic fertilizers and chemical plant protection measures, a significant loss of soil microbial (bacterial and fungal) diversity has been recorded as well [11]–[13].

Further, in most regions of Europe, legal regulations have (until recently) outlawed the sale of local agrobiodiversity contained in indigenous or local, non-uniform seeds. For decades, preference has been given to uniform, certified seeds that guarantee high yield, but rely on external support (in the form of chemical fertilizers and pesticides) for consistent performance [14]–[16].

With increasing environmental concerns, policy makers are facing growing consumer demand for organic and local food [17, 18]. Accordingly, regulators are seeking means of enhancing farmer migration to organic farming, particularly farming that uses local agrobiodiversity.

Responding to consumer demands and evolving scientific understanding of the importance of agrobiodiversity for sustainable agriculture, the new EU Organic Regulation (EU 848/2018) was adopted in May 2018. It seeks to promote the use of agrobiodiversity (referred to in the legislation as "heterogeneous materials"), in organic agriculture. The regulation permits such materials to be marketed without the need to pass cumbersome and costly seed certification requirements under existing laws. It also acknowledges that "there could be benefits of using such diverse material... to reduce spread of diseases, improve resilience and increase biodiversity."

Yet, decades of disengagement with local agrobiodiversity and associated farming practices has led to a situation where the mere permission to use and sell "heterogeneous materials" may not be adequate to trigger changes in farmers' choices. Concrete incentives, including economic incentives and reeducation efforts may be needed to accomplish this goal [19]– [21].

Further, gaps in current scientific knowledge on best practices for profitable and sustainable use of agrobiodiversity [22, 23], make farmer migration to organic farming a risky endeavour. Current scientific understanding can profit greatly from on-the-ground experience of farmers (e.g., small farmers and indigenous communities) that are actively engaged in agriculture with local agrobiodiversity. However, an equitable, transparent and trustworthy means of incentivizing and facilitating farmer-sharing of local know-how and materials does not currently exist.

The present paper investigates whether, and the extent to which a blockchain-based solution could contribute to overcoming existing regulatory and practical hurdles and incentivize in situ CRI with agrobiodiversity.

Further, We find that blockchain technology may empower data providers, including small farmers, to collectively track, control and monetize the usage of data and assets shared, while minimizing illegal transactions. Transaction costs may also be lowered by removing complex and expensive interaction processes. However, blockchain solutions primarily deal with linear flows of data and information. In order to operationalize a B4A usecase in the real world, specific technological addons may be necessary that permit multiple entry and access points for data, know-how as well as associated materials (e.g., seeds). Further, unlike bitcoins that can be traded only once, farmers' knowhow (and materials) can be simultaneously traded with multiple "buyers" on the chain. Also, the B4A usecase is quite unlike a blockchain application for tracking agricultural supply chains, where the underlying materials are not significantly transformed post transfer from one stakeholder to the next. In the B4A usecase, the traded material (seed, soil samples, know-how) will get significantly transformed over time: This is desirable, but makes blockchain based traceability and benefit sharing challenging. Hurdles associated with physical materials transfers and stakeholder digital identities can be resolved by complementary development of biomarkers and by implementing trust checkpoints, respectively. However, further research and development is necessary to design an ethical and sustainable blockchain based solution to incentivize in situ conservation, research and innovation with agrobiodiversity. Some future directions of research are recommended. Index Terms—Agrobiodiversity, Blockchain, Incentives, Sustainability, Access and benefit sharing I. INTRODUCTION

Biodiversity is defined as "the variability among living organisms from all sources . . . and the ecological complexes of which they are part; this includes diversity within species, between species and of ecosystems" (CBD). The broad term biodiversityincludes terrestrial and aquatic biodiversity, as well as agricultual biodiversity (agrobiodiversity) [1]. This article focuses on agrobiodiversity. Agrobiodiversity comprises of diversity within the soil (the soil microbiome) and on the soil (genetic diversity contained in indigenous or local, nonuniform seeds, landraces, pollina- tors etc.). Efforts to conserve agrobiodiversity in its natural surroundings, such as through farming activities, is called in situ conservation. Emerging research reveals that farming systems that enhance agrobiodiversity are often more robust in the face of local biotic and abiotic stresses [2]. Agrobiodiversity, therefore, actively contributes to food and nutritional security in the face of rapid climate change [2]-[4]. International instruments, such as the Convention on Bi- ological Diversity (CBD) [5] and the International Treaty on Plant Genetic Resources for Food and Agriculture ("the Seed Treaty") [6], emphasize the relevance of in situ agro- biodiversity conservation. They also require member states to adopt so called "Access and Benefit Sharing (ABS)" regimes. ABS regimes aim to facilitate research and innovation with agrobiodiversity on one hand, and grant monetary benefits (so called "benefit sharing"), to those granting access to their bioresources. However, research reveals that due to several regulatory failures, the ABS system has failed to optimally incentivize and promote in situ conservation, research and innovation (CRI) with agrobiodiversity [8, 9]. Since the 1900s, an estimated 75 per cent of crop genetic diversity has been lost [7] due, inter alia, to expanding farm area under commercially available uniform seeds. As a consequence of the intense use of inorganic fertilizers and chemical plant protection measures, a significant loss of soil microbial (bacterial and fungal) diversity has also been recorded [8]-[10]. Further, in most regions of Europe, legal regulations have (until recently) outlawed the sale of local agrobiodiversity contained in indigenous or local, non-uniform seeds. For

The paper is arranged as follows: Following this introduction, Section II provides a sketch of the background leading up to the research questions guiding this paper. More specifically, it summarizes the findings from extensive empirical research conducted with the aim of understanding (i) whether existing intellectual property rights regimes and associated policies incentivize in situ conservation of agrobiodiversity, and (ii) the needs of farmers currently engaged with in situ agrobiodiversity conservation and improvement. Section III then identifies the research questions that guide this paper. With the help of an exhaustive literature review, section IV identifies hurdles in international instruments contributing to sub-optimal in situ CRI with agrobiodiversity.

Previous work has recommended using blockchain technology to incentivize CRI with agrobiodiversity [21]. Section V summarizes the key features of blockchain technology with a view to providing an overview of reasons why B4A may be an appropriate blockchain usecase. Section VI goes deeper into this question: section VI A tabulates and categorizes the regulatory hurdles, lacunae and farmers' needs identified in sections II and IV, to determine which of them can, and which cannot, be addressed by blockchain technology. Section VI B focuses on the issue of incentives: emerging research suggests that on chain and off chain incentives can be combined to better accomplish real world results. Incentives that can be built in and off chain to support the cause of in situ CRI with agrobiodiversity are explored. Section VII concludes with recommendations for future lines of research that can support blockchain or Distributed Ledger Technology (DLT)¹ based incentivization of in situ conservation, innovation and research with agrobiodiversity by all stakeholders.

II. BACKGROUND AND RESEARCH QUESTIONS

A. Intellectual Property Protection Regimes, Inequitable Incentive Structures and Market Failures

This article builds on extensive legal and empirical research conducted to determine whether existing intellectual property protection laws and associated governmental policies are able to promote and incentivize CRI with agrobiodiversity conserved in situ [8, 9, 19]–[21, 24]. Arguably, conservation is not innovation. However, research conducted from multidisciplinary, legal as well as ecological economics approaches have recommended that for 'sustainable innovations'² in 'plant varieties'³, recognizing farmers' efforts as 'innovation', and not merely as 'conservation' is necessary and can support the cause of CRI with agrobiodiversity [20, 25]–[28]. Recognizing farmers' contributions as innovations can also help address issues of growing rural-urban migration (NO REFER-ENCE FOUND) and growing disinterest in agriculture among younger generations of farmers (NO REFERENCE FOUND).

In existing literature, seed sector innovators are broadly classified into two groups [29]: (i) formal innovators, i.e. plant

¹Blockchain is a subset of Distributed Ledger Technologies. For simplicity, we use the term *blockchain* but also include technologies that are not based on a block structure.

²Kochupillai, 2016 was perhaps the first to use and define the term "sustainable innovations" as innovations that: (i) Protect natural resources, inter alia, by supporting in situ conservation and improvement of agrobiodiversity. (ii) Facilitate both formal and informal (downstream) innovations to continue to take place in the generations to come. (iii) Equitably incentivize participation by all potential innovators in the process (life cycle) of innovation.

³Kochupillai's research deviated from the international legal understanding of the term 'plant variety'. The term "plant varieties" in Kochupillai's original work includes 'agrobiodiversity' as well as innovations on and with this diversity by breeders ("the formal sector") as well as farmers ("the informal sector"). This deviation was necessary because the Indian PPVFR Act, 2001, which was the main legislation Kochupillai's research focused on, includes not only breeders' varieties, but also so called "extant varieties" and "farmers' varieties" within its scope. Farmers' variety, has also been defined in the Indian legislation to include traditionally cultivated varieties, as well as wild relatives or landraces of a variety. Kochupillai's research focused on India for three reasons: India is a major (agro)biodiversity hotspot, almost 60 per cent of India's population relies on agriculture for its livelihood, and the Indian Protection of Plant Varieties and Farmers Rights Act, 2001 (PPVFR Act) is considered to be one of the most unique in the world as it purports to accomplish a complex three-fold goal: (i) promoting private sector innovations in plant varieties, (ii) protecting and promoting farmers' innovations in plant varieties, and (iii) rewarding farmers and farming communities engaged with conserving agrobiodiversity.

breeders affiliated with Universities, research institutions or the seed industry, and (ii) informal innovators, i.e. farmers (particularly small and marginal farmers, who constitute almost 80 per cent of India's farming community).

A growing body of literature highlights the critical relevance of in situ conservation of seeds by farmers (the informal innovators). Particular relevance is placed on the saving of local, genetically variable seeds, that house diverse genetic materials necessary resisting pests and diseases, and ensuring food security in the face of climate change [30]–[32]. Agrobiodiversity conserved in situ, is also an important raw material necessary for downstream research and plant breeding by the formal sector.

Accordingly, researchers have argued that legal regimes that sub-optimally incentivize saving of local, genetically variable seeds, also undermine the continuation of plant breeding and seed improvement in the long run [19]. Kochupillai's legal and empirical research in India also identified several legal and policy hurdles that act as "perverse incentives", actively disincentivizing seed-saving.

Existing incentive structures have also been found to be either inadequate or inappropriate to incentivize CRI with agrobiodiversity. Intellectual property laws and associated governmental policies have also been found to inequitably skew the innovation landscape by (i) promoting only formal innovations using existing proprietary germplasm stores (e.g. in seed banks), and (ii) neglecting innovations that rely on and promote in situ agrobiodiversity conservation [19, 21], and CITE OTHERS).

B. Incentivizing Sustainable Seed Innovations: Challenges faced by Farmers

Past research has called for urgent legal and regulatory attention to level the seed-innovations landscape, and to promote sustainable innovations in plant varieties [?, 24]. Accomplishing this complex goal involves a rebalancing of incentives, roles and contributions among diverse stakeholders operating in a delicate ecosystem.

To find "means of incentivizing farmer-level innovations on and with seeds from often-neglected indigenous varieties in India", and, in particular, with the "greater role that public recognition for (small) farmers' seed innovation might play therein", a research grant was acquired from the UK Arts and Humanities Research Council (AHRC) in 2016-17.

The grant supported further empirical research and brought together experts and farmer-stakeholders to a one-day workshop designed to intensively debate and elicit diverse views on this topic. The workshop, hosted by the AHRC funded partner, the Art of Living Foundation, an international NGO working to revive traditional indigenous seeds, associated farming systems ("natural farming") and know-how, created 4 working groups. ⁴ All groups identified current challenges that either disincentivize or prevent active adoption of indigenous

⁴The working groups focused on law, research, outreach and awareness and participatory plant breeding, Moderat by Sunita Sreedharan, Mrinalini Kochupillai, Shamika Mone and S.C. Tripathi respectively.

seeds (agrobiodiversity) by small and marginal farmers, and made recommendations for future actions to overcome these challenges.

The challenges and recommendations emerging from the workshop were divided into the following 10 broad categories (ι_i) [21]:

- ι_1 Inadequate attribution and recognition: As indigenous seeds are heterogenous and not uniform, once sold or exchanged, it can be difficult to identify the next generation of seeds as originating from a specific source. Farmer-sellers of high quality indigenous seeds, therefore, never get the recognition and attribution they deserve. This finding has also been confirmed by other researchers [24, 33, 34]. Farmers also emphasized the importance of giving unique names to seeds developed by farmers, and of acknowledging inventor-communities and persons engaged with farming using agrobiodiversity and traditional know-how. Recognition must also be given to those who make local improvements to indigenous seeds and knowhow, especially when these improvements result in higher yields or better quality of produce.
- ι_2 Sub-optimal economic and other incentives: As a result of (i) inadequate (or no) attribution, (ii) low demand for indigenous seeds and associated know-how, and (iii) unavailable markets and government support in the form of a "minimum support price" for the produce resulting from unique local seeds and grains, innovative farmers who cultivate with and improve agrobiodiversity in situ, do not have optimal incentives to continue this work. While farmers (in India) are happy to share their materials and know-how with other farmers, such other farmers have no concrete incentive to migrate to farming systems that use indigenous, heterogeneous seeds and improve agrobiodiversity.
- t_3 Traceability: There are currently no mechanisms in place that can track and trace the source of indigenous seeds and of associated know-how. This prevents the equatable and honest sharing of benefits on one hand, and the emergence of markets selling high quality indigenous seeds on the other.
- ι_4 Sub-optimal research and education: As farmers often do not have the know-how necessary to beneficially use indigenous seeds, research and education (formal and informal channels), including through community engagement is centrally relevant to promote adoption of farming techniques that utilize, conserve and improve agrobiodiversity in situ.
- ι_5 Reliable communication channels (e.g., to report problems, success stories etc.) are needed to facilitate communication between farmers and others engaged with agrobiodiversity.
- ι_6 Marketing channels and new markets are necessary to facilitate quick sales of produce and seeds resulting from

using indigenous seeds.⁵

- ι_7 Means of ensuring and documenting quality of seeds/produces and accessibility of documentation.
- ι_8 Means of ensuring quantity, timely availability, and affordability of indigenous seeds and access to know-how on when and how to sow and cultivate them.
- ι_9 More Knowledge creation/verification, including through research in various disciplines and dissemination of research findings; currently there is sub-optimal research being conducted on characteristics and strengths of indigenous seeds and associated traditional agricultural systems.
- ι_{10} Revival/Maintenance of Traditional Ecological Knowledge (TEK) and local cultures associated with it.

C. Recommending a Three-Pronged Approach

With the aim of making more specific recommendations for concrete policy and regulatory action based on the key recommendations of the working groups, an impact acceleration grant was sought and acquired from the UK Global Challenges Research Fund (2019). The position paper for the Government of India, compiled using the GCRF grant, recommended a three pronged approach for incentivizing sustainable indigenous seed innovations [21, 35]:

- 1) Prong 1: Reviving Traditional Ecological Knowledge Systems, that contain rich knowledge on means of protecting and enriching in and on soil biodiversity
- Prong 2: Re-designing Educational Curricula of Agricultural Universities and of Rural Agricultural Extension Services to incorporate extensive education and training in farming systems that incorporate this traditional ecological knowledge; and
- 3) Prong 3: Re-leveling the Incentives Landscape, inter alia, by adopting technical solutions such as blockchains to overcome hurdles that currently disincentivize research and in situ innovations with agrobiodiversity by farmers and researchers (i.e. formal and informal sectors).

This article focuses and builds on the third prong and is guided by the following research questions:

III. RESEARCH QUESTIONS AND METHODOLOGY

A. Research Questions

- **RQ 1a.** Which major international regulatory regime(s) are responsible for promoting in situ agrobiodiversity conservation and its equitable access, use and benefit sharing?
- **RQ 1b.** Which shortcomings, if any, in these regimes, may be contributing to sub-optimal in situ conservation, research and innovation with agrobiodiversity?
- **RQ 2.** Is the promotion of research and innovation with agrobiodiversity conserved in situ an appropriate blockchain use case?

⁵it is necessary to note that some farmers' groups expressed opposition to the idea of seed "sales" and preferred to give away seeds and associated know-how for "free". However, all were interested in creation of appropriate channels for transfer of seeds and know, whether it be for free or for a charge

- **RQ 3.** How, and to what extent, can a blockchain-based solution help:
 - RQ 3a. address the identified shortcomings and challenges (RQ 1b.), and
 - RQ 3b. provide incentives for farmers to use, innovate with, and share traditional know-how and agrobiodiversity conserved in situ,

In relation to RQ 3a, the article looks into current capabilities and shortcomings of blockchain solutions that can help overcome challenges and implement recommendations identified by Kochupillai 2016, Kochupillai et al. 2019, and under RQ 1.

B. Methodology

This paper provides insights into the above research questions based on a comprehensive literature review. The search terms used for the literature review, as well as the outlines of the blockchain "Ecosystem and Conditions" are guided by the insights and recommendations gathered from the background research described in Part II above.

IV. THE INTERNATIONAL REGULATORY REGIME FOR THE CONSERVATION AND EQUITABLE USE OF AGROBIODIVERSITY

A. Overview of the Major International Regulatory Regimes (RQ1a)

In the late 1980s, researchers found that biodiversity declined to "its lowest level since the end of the Mesozoic, 65 million years ago" [36]. With growing civil and political awareness about the rapid decline of global biodiversity, the need for an international legal instrument to protect both wild and domesticated diversity was voiced [37]. Eventually, the Earth summit of Rio in 1992 put the topic on the political agenda and collected signatures for the adoption of the Convention on Biodiversity (CBD) [6, 38].

The CBD has three main goals, namely (i) the conservation of biodiversity, (ii) the sustainable use of biodiversity and (iii) the fair and equitable sharing of benefits resulting from the use of biodiversity and associated plant genetic resources (PGRs) [39]. The exchange of PGR through the CBD is based on national bilateral "case-by-case negotiations" [40]-[42], whereby national rules govern the contracts [38, 43]. In spite of the stated goals, PGR were barely exchanged in the first twenty years of the CBD's implementation, suggesting a need to review and amend its provisions [44]. The CBD was extended in 2014 by a supplementary agreement, the Nagoya Protocol (NP) [45], which aimed, inter alia, to overcome shortcomings connected to the CBD's bilateral system. The NP aims to incentivize exchange of PGRs and consequently the promotion of in-situ conservation and engagement with PGRs. The NP clarified previous terminological ambiguities and added an article to facilitate access and benefit sharing (ABS) of PGRs.

In 2004, a further international regulation aimed at protecting agricultural biodiversity, was adopted. Titled the International Treaty on Plant Genetic Resources for Food and Agriculture (ITPGRFA), it was implemented by the Food and Agricultural Organization of the United Nations [46]. Also known as the Seed Treaty, the ITPGRFA aims to facilitate access to a wider range of PGRs (particularly those relevant for agriculture) through a central sharing point [44, 47, 48]. The Seed Treaty replaced bilateral negotiations with a multilateral system, which includes a Standard Material Transfer Agreement (SMTA) [49] for essential food and fodder crops [50]. The SMTA frames responsibilities and duties for providers and users of PGRs, including monetary and physical sharing in case of commercial usage of PGR ("benefit sharing") [38, 51]. The ITPGRFA is considered "the most sophisticated international benefit-sharing mechanism to date" [38].

Although the CBD and the ITPGRFA have been ratified by most countries, PGRs were barely exchanged [44, 47, 48]. In fact, studies find that the NP has complicated the access to PGRs, especially in biodiversity hotspots such as South Asia, East Africa, and South America [48, 52]. The limited use of PGR under the NP has also been linked to the small number of countries that facilitate access to PGRs through additional legal or administrative mechanisms which implement the CBD/NP.

PGRs are accessed more frequently under the ITPGRFA, however, research institutes struggle with several barriers and bureaucratic hurdles when accessing PGRs [8, 9, 44]. The ITPGRFA aims to protect incentives for sharing,e.g., by not including intellectual property rights for PGRs in its treaty [53]. To better understand the specific strengths and shortcomings in the international regulatory regimes, the following literature review was conducted (RQ 1b, below).

B. Challenges and Shortcomings of International Regulatory Regimes (RQ1b)

To answer RQ1b, the current research on the topic was analysed through a systematic literature review and analysis of the relevant results. The literature review was conducted on the ISI Web of Knowledge database, particularly two subdatabases therein: (i) Current Contents Connect (=database 1, DB1) and (ii) Web of Science Core Collection (DB2). The search was carried out in September 2020 for publications in the English language from 1998 to 2020. The screening followed a three-step approach:

Step I: A broad keyword search was conducted. The keyword search in Step I looked for titles including the term "biodiversity", "access and benefit sharing" and "agriculture". *TS*=((*biodiversity AND Access and benefit sharing AND agriculture*).

The structured literature review yielded (i) 161 results in DB1 and (ii) 230 results in DB2. The topic of agrobiodiversity conservation and equitable access and benefit sharing is primarily covered under two international regulatory regimes (see section IVA above) [6, 38], namely, the CBD and its extension, the Nagoya Protocol (NP) and the (ii) International Treaty on Plant Genetic Resources for Food and Agriculture (ITPGRFA). As these regimes operate at the global level and almost all nations of the world are signatory to these [19] and the shortcomings of these regimes were the focus of RQ1b, for

step II, papers were eliminated that did not mention at least any one of the two regimes.

Step II: Step 1 combined with *TS*=(*objectives AND Nagoya Protocol OR International Treaty on Plant Genetic Resources for Agriculture OR ITPGRFA*)

Step II yielded 38 papers in DB1 and 62 papers in DB 2 providing detailed information on the two regimes. We screened the papers to identify the key strengths of the two regimes. The findings are displayed in table I.

 TABLE I

 Strengths of existing international biodiversity protection

 REGIMES (NAGOYA PROTOCOL (NP)/CONVENTION ON BIODIVERSITY

 (CBD) and the International Treaty for Plant Genetic

 RESOURCES FOR AGRICULTURE (ITPGRFA)) identified within

 LITERATURE

Strengths σ_i	NP/CBD	ITPGRFA
σ_1 : Integration of ethical and	[54, 55]	[54, 56, 57]
legal principles in ownership	[56, 58]	
and usage of PGR	[59]–[62]	
σ_2 : Objectives comprise	[59, 62]	[57, 61, 63]
indigenous knowledge		
σ_3 : Following "noble	[62, 64]	[57, 58, 60]
objectives"		
σ_4 : Ratified by most countries;	[65, 66]	[57]
inclusive		

Thereafter, these international treaties were scanned to extract and study the provisions that deal specifically with agrobiodiversity conservation, use, access or benefit sharing. Under various provisions, the treaties underscore the key role played by indigenous communities and their traditional knowledge in safeguarding and bringing forward (agro)biodiversity from generation to generation [67, 68]. Accordingly, these treaties emphasize and mandate benefit sharing with communities that grant access to their biodiversity, traditional knowledge and know-how for downstream use (including for use in research or for commercial purposes) (see Nagoya protocol Article 14-16; Article 10.2 of the ITPGRFA).

Beyond ensuring a "fair price" for these valuable resources, the goal of the benefit sharing regime is twofold: (i) to ensure that conservers of (agro)biodiversity, have the incentives and monetary means to continue their good work [54], and (ii) to give conservers of biodiversity concrete incentives to share (give access to) their resources and know-how with other stakeholders in the value or innovation chain [67, 68].

Indeed, agrobiodiversity and associated Plant Genetic Resources (PGRs) are actively sought by scientists engaged in sustainable agricultural research [22, 69]. However, due to the link between agrobiodiversity and exchanged PGRs, the effectiveness and the achievement of goals of agrobiodiversity regimes is validated best by analysing Access and Benefit Sharing (ABS) agreements [70]. Following different approaches, both regimes have only witnessed very minimal success in promoting equitable PGR access and benefit sharing. Under the Nagoya Protocol, on an average, 2.05 ABS agreements per year, per country have been signed [44, 50]. Under the ITPGRFA, from more than 3.3 million samples exchanged, no benefit-sharing occurred [70]. To find the

TABLE II

SHORTCOMINGS OF EXISTING INTERNATIONAL BIODIVERSITY PROTECTION REGIMES (NAGOYA PROTOCOL (NP)/CONVENTION ON BIODIVERSITY (CBD) AND THE INTERNATIONAL TREATY FOR PLANT GENETIC RESOURCES FOR AGRICULTURE (ITPGRFA)) IDENTIFIED WITHIN LITERATURE

Shortcomings	NP/CBD	ITPGRFA
ι_{11} :Lack of traceability of	[48, 73, 74]	[58]
PGRs exchanged		
ι_{12} :No access to PGRs due to		
- $\iota_{12,1}$: Lack of competences	[75]	[8]
- $\iota_{12,2}$:Delays and stoppages	[9, 70, 74, 76]	[8, 50, 58, 77]
- $\iota_{12.3}$:Incomplete catalogues	[55, 78, 79]	[44]
of PGRs		
- $\iota_{12.4}$:Legal uncertainties	[44, 59, 64, 78,	[70]
_	79]	
- $\iota_{12.5}$:Usage of "irritating,	[62]	[72, 73, 79]
vague terminology"		
- $\iota_{12.6}$:High transaction costs	[9, 74, 80]	[62, 70]
- $\iota_{12.7}$:Lack of trust	[81, 82]	[70]
ι_{13} :No rules for sharing digital	[44, 83]–[86]	[66, 87, 88]
sequence information (DSI)		

shortcomings, hurdles, problems and weaknesses of existing regimes, in step 3 of the literature review, papers that discuss "shortcomings" of these regimes were extracted and analysed.

Step III: Step 2 combined with *TS*=(*shortcomings OR problems OR hurdles OR limitations*)

Step III found 10 papers in DB1 and 11 papers in DB2 reporting that despite good intentions of international treaties, cumbersome bureaucratic and regulatory hurdles therein delay and disincentivize honest research practices. They thereby prevent equitable benefit sharing, creating an atmosphere of mistrust between providers and users of agrobiodiversity [71]. Lack of trust leads to sub-optimal accessibility and benefit sharing, triggering a vicious cycle where rampant biopiracy further reduces trust among stakeholders [72]. Table II lists shortcomings predominant in literature.

Current regimes are inadequate or unable to (i) trace PGRs to source, (ii) identify and penalize infringements, and (iii) incentivize access and benefit sharing. They thus fail to facilitate and incentivize legitimate and honest access to and research with PGRs. Further, they currently fail to integrate digital sequence information (DSI) associated with PGRs in their access and benefit regulations. Despite their significant role in research and development [89], DSI currently bypasses any ABS and obligations or values thereof [84]. Both the CBD/NP and the ITPGRFA have recently tried to include DSI, but describe the struggle to accomplish this as being a result of "pre-existing weakneeses" in their design [86].

The ongoing discussions on DSI also emphasize the need to rethink existing administrative and regulatory frameworks and to identify alternative or complementary solutions. Such solutions should facilitate the tracing and tracking of shared data to its source and to every downstream use [54]. It should also sufficiently protect and incentivize the maintenance of the source of PGRs and DSI [84, 86]. Externalising transaction costs associated with ABS and fostering interdisciplinary and cross-cultural research with PGRs are also worthwhile goals to pursue [44].

Regulatory and bureaucratic delays result in researchers avoiding regulatory check posts [77, 79], blurring the origins of their data, or stopping research out of fear of allegations of biopiracy [44, 48, 59, 73]. As a result, international regulations that aim to protect, conserve and equitably regulate the access to (agro)biodiversity [85], currently limit the availability and access to valuable PGRs, and inadvertently disinentivize honest use of the systems established under these regulations (For the ITPGRFA: [42, 50]. For the NP/CBD: [8, 48, 52, 90]. Both regimes are currently looking for solutions [54, 66, 70, 84, 86].)

V. BLOCKCHAIN USE CASES: FEATURES AND CONDITIONS (RQ2)

A. Blockchain: Overview of Key Features

Blockchain as a technology was first applied in the decentralized cryptocurrency Bitcoin in 2008 [91]. As first application, Bitcoin is used for pseudonymous payments without the need to rely on a trusted third party. Since Bitcoin increased in popularity and communities understood that a blockchain network can be utilized for other purposes or use cases, applications outside the financial sector arose [92].

Most scientific literature [93]–[97] highlights four key advantages of blockchain that make it suitable for diverse use cases:

- Trustlessness: Parties who do not fully trust each other are able to interact with each other. This absence of trust is the result of absolute transparency [98]. Each action of a user is visible to and verifiable by all other participants. Furthermore, it is non-repudiable, such that the acting user can never argue that a particular action did not happen.
- 2) Independence from third parties (or intermediaries): The trustlessness of the system comes with the fact that the parties do not need a trusted third party to run the system. Instead, the members of the network run the system together, without a single party being able to manipulate the state. Only if a specific threshold of accomplices is exceeded, manipulation is possible.
- 3) Automation and Execution: The maintenance of a consensus, especially about the evolution of a set of shared facts, is crucial. It allows the parties of the network to define rules to which everyone has to adhere; otherwise, participation is restricted, or the respective parties are punished for their malicious behavior. These rules are either a core part of the consensus mechanism or can be formulated in so-called Smart Contracts; little programs installed on the blockchain, running as autonomous services. With these Smart Contracts, these rulings can be extended to every aspect of the network, allowing participants to collectively define processes, responsibilities, roles, and outcomes.
- Built-in incentive mechanisms: Incentives have been a key pillar in blockchain solutions. The incentive mechanisms existing in Bitcoin have lead to huge energy

consumption and carbon footprint [99, 100], innovation in hardware and a market capitalization of over 280 billion USD in under 13 years of existence [101]. Further networks and applications have pulled many stakeholders into their ecosystems by proper incentive mechanisms. These incentive mechanisms usually have a two fold approach: a) reward good behavior and b) punish bad behavior.

Therefore, data platforms based on permissioned public or consortium blockchains with appropriate software architectures and governance models can help solve problems of (lack of) trust, traceability, and equitable data collection [102]. They empower data providers to collectively track, control and monetize the usage of their contributed data and assets. It also allows participants (farmers' groups, for example) to collectively decide the terms under which they will transfer materials, know-how and other data.

These features of blockchain appear to make it a useful technology for overcoming the identified challenges and shortcomings and incentivizing in situ conservation and use of agrobiodiversity. In the following sub-section we look more closely at whether and what extent, blockchain may be a good technological fit for this use case.

B. Blockchain for Agrobiodiversity (B4A): An appropriate Blockchain Use Case?

Blockchain applications in the agro-food sector are currently dominated by track and trace solutions [103, 104], aimed primarily at facilitating quality control and provenance checks [105]. Other applications permit direct sales from farmers to consumers [106], removing intermediaries [107], expediting payments with the help of smart contracts and enhancing farmers' profits from direct sales of products to consumers [108]. Emerging research on blockchain and incentives, however, suggests that features of blockchain technology, especially when used in combination with other emerging technologies, e.g., Internet of Things (IoT) [109, 110], make it potentially beneficial for applications that go beyond supply chain tracing and automated payments [106]. In particular, blockchain can be used to incentivize activities that existing regulations (e.g., due to regulatory and market failures), have been unable to incentivize [111].

Blockchain technology is particularly appropriate when one is dealing with

- 1) transactions, especially a chain of continuous, inter-linked transfers of valuable assets from one stakeholder (or buyer) to the next [112],
- 2) a use case that values or requires immutable record keeping [113]
- 3) a use case that has multiple stakeholders that do not trust each other [97],
- 4) a use case that requires "checkposts" that can check for data veracity or reconcile disparate data [114], and
- 5) a use case that values de-centralized rather than centralized management and monitoring of the system [115].

In the B4A use case, the features required to overcome the identified challenges correspond with all of the above points that indicate an appropriate blockchain use case. Specifically, as discussed in parts II C and III above, B4A requires

- Corresponding with 1) above, facilitation and monitoring of transactions involving the transfer of agrobiodiversity in the form of heterogeneous seeds, soil samples, and associated know-how (e.g., know how on best practices to use/cultivate these seeds, or the specific characteristics of these seeds);
- Corresponding with 2) above, immutable record keeping in order to be able to trace the original source from which these valuable materials and know-how started being transacted - this is necessary, inter alia, to permit attribution and benefit sharing;
- Corresponding with 3) above, managing and incentivizing multiple stakeholders that do not trust one another or do not trust the existing centralized system
- 4) Corresponding with 4) above, "check-posts" (or nodes) that have incentive to research and verify the quality, characteristics (of materials) and authenticity (of knowhow/information) transacted - this would incentivize research on and with agrobiodiversity (see Part II C above); and
- 5) Corresponding with 5) above, a distributed system, that would facilitate more "transactions" (or information sharing) without having to rely on a single/central authority, thereby creating more communication and marketing channels (see Part II C above) for heterogeneous seeds, and associated know-how and DSI (where applicable).

The following section VI discusses the current state of art in blockchain technology and the extent to which the technology can help overcome the challenges identified in sections II and IV by incorporating the above features.

VI. BLOCKCHAIN FOR AGROBIODIVERSITY: POSSIBILITIES AND LIMITATIONS (RQ3)

A. Issues that Blockchain can and cannot help addressing

Research question 3a challenges us to analyze if blockchain helps us to address shortcomings and challenges identified in section II and IV. To understand whether, and to what extent, blockchain is an appropriate technological choice for incentivizing conservation, use and research with agrobiodiversity, we first outline the possible scenarios vis-a-vis impact of technological deployment. This has been done in terms of (possible) influence on any of the identified challenges and shortcomings in the existing systems. Specifically, four scenarios are possible:

- **Blockchain solves the problem** (++): An issue can be directly solved by blockchain. Depending on the complexity, blockchain might help by overcoming issues of communication, provenance or data integrity.
- Blockchain supports other approaches in solving the issue (+): Blockchain itself might not be the sole solution, but might enable other approaches which utilize the

TABLE III CATEGORIZATION OF ISSUES AND SHORTCOMINGS

Issue/Shortcoming/Problem	BC Impact
Farmer Challenges	
ι_1 Importance of attribution	++
ι_2 Economic and other incentives, royalty and rewards	++
ι_3 Traceability of source of seeds and know-how	++
ι_4 Research and education	+
ι_5 Communication channels	+
ι_6 Marketing channels for sales of produce and seeds	+
ι_7 Means of ensuring and documenting quality of seeds	++
ι_8 Means ensuring quantity, availability and affordability	++
ι_9 Knowledge creation and verification	++
ι_{10} Perpetuation of traditional ecological knowledge	+
Shortcomings identified in Literature Review	
ι_{11} Lack of traceability of PGRs exchanged	++
ι_{12} No access to PGRs due to	
- $\iota_{12.1}$ Lack of competencies	+
- $\iota_{12,2}$ Delays and stoppages	+
- $\iota_{12.3}$ Incomplete catalogues of PGRs	++
- $\iota_{12.4}$ Legal uncertainties	0
- $\iota_{12.5}$ Usage of "irritating, vague terminology"	0
- $\iota_{12.6}$ High transaction costs	++
- $\iota_{12.7}$ Lack of trust	++
ι_{13} No rules for sharing digital sequence information	++
ι_{14} Political pressures	0
ι_{15} Policy-based imperatives	0
ι_{16} Inequality, powerbalances changing at community level	-

blockchain in solving the issue. This is often the case if blockchain and the data stored on it is used for other purposes, e.g., data analysis, research or education.

- Blockchain has no impact on the problem (o): Even if implemented, problems such as mindsets and systemic roadblocks will not (automatically) be solved or impacted by blockchain. In the face of such issues, it is irrelevant whether blockchain is implemented or not. However, resolving these issues (e.g. through education and prioritization under policy/regulation) can pave the way for a solutions based on blockchain.
- Blockchain negatively impacts the problem(-): Some problems might get elevated by a blockchain-specific implementation. Properties such as immutability and inability to delete data might conflict with other goals of a project, such as GDPR or data privacy, even putting lives of people/farmers at risk.

We sort aforementioned issues, shortcomings and problems from the literature and attribute them a category (++, +, oor -). All items are depicted in Table III. This allows us to focus on the issues that can be solved by blockchain and evaluate whether the current state of blockchain is sufficient to address these problems. As the issues of incentives is one that is particularly complex in the B4A usecase, we discuss it separately in the following sub-section.

1) Issues that may be supported by Blockchain: From Table III it becomes clear that issues or categories of issues that relate to communication and knowledge transfer and preparation, e.g., research and education (ι_4), communication channels (ι_5), marketing channels for sales (ι_6) or perpetuation of traditional ecological knowledge (ι_{10}), can benefit from the ecosystem created by a blockchain network. Within

this ecosystem, additional services can be created (e.g. in the form of distributed apps) that serve diverse purposes of the stakeholders (especially farmers) in the network, without directly relying on blockchain technology. It is imaginable, for example, that other stakeholders (e.g., research institutes) offer specialized data analysis or research based verification based on the data, or know-how available on-chain.

In fact, the compilation of data and know-how from diverse sources via the B4A system would facilitate and enhance research on and with agrobiodiversity. B4A could facilitate the identification of relevant material, if, for example, search functions supported by AI or ML models are built on it. B4A would also support speedy and traceable transactions of the material and associated know-how. However, it is noteworthy that blockchain cannot help identify materials once they are removed from tamper proof packaging and are transformed. Therefore, blockchain technology will need support from other technologies such as biomarker technology to support tracking and tracing throughout the seed innovation lifecycle.

2) Issues that may be aggravated by Blockchain: Some issues are not affected, or are negatively affected by blockchain: Especially in the relationship with the government and the larger society, problems might be elevated. Managing data or incentive structures over a blockchain network that requires governmental or legislative action ($\iota_{12.4}$) for proper functioning, might lead to delayed or competing actions. Approaches to reduce inequality or re-balance economic power structures at a community level might lead to counter-movements or, in the worse cases, to violence and civil unrest (ι_{16}).

3) Issues that may be addressable by Blockchain: We divide the issues that may be addressable by blockchain, into three distinct categories: ν_1) Identity, attribution and non-repudiation, ν_2) provenance, tracking, and information verification ν_3) disintermediation and cost reduction (See table IV).

Although a blockchain solution can potentially provide these features, there are important limitations resulting from the complexity of the B4A use case, including the number of stakeholders involved, the need for diverse smart-contract terms from each participating farming-community/ contributor of materials and know-how, as well as the diversity of materials and data that needs to be transacted. Accordingly, in the following sub-section, for each of the categories, we describe a) the state of the research and b) the current limitations that need to be overcome in the context of B4A as well as recommendations to approach the issues.

B. Possibilities and Limitations under the Current State of Blockchain Art

1) ν_1 : Identity, attribution and non-repudiation: The reliable creation and management of identities is highly relevant in every blockchain application [116]. A decentralized identity management allows the creation of arbitrary amounts of identities without requiring a third party. Furthermore, messages that are signed with these identities are tamper-proof and non-rebuttable.

The creation of such identities relies on asymmetric cryptography [117]. Often, key-derivation algorithms are used to generate private and public keys generated randomly by the computer. However, in the context of B4A-use case with realworld attribution, two key challenges remain:

- How is an identity that exists in the blockchain linked to a real-world entity, e.g. a farmer or institution? As our use case includes multiple stakeholders with different roles and responsibilities within the system, we need to be able to connect real-world entities with on-chain identities. Otherwise, we cannot distribute respective rights and roles to the stakeholders in the system.
- How can we prevent Sybil-attacks (multiple creations of identities)? As regular blockchain solutions allow the creation of an arbitrary amount of digital identities, we need to ensure that only one real-world identity controls one on-chain identity with respective rights. Otherwise, one party could create multiple accounts acting on behalf of many.

The complexity of this problem is further increased as several of our key stakeholders a) are not in possession of standardized forms of digital identities and b) can be legal or natural persons, requiring diverse approaches for establishing identities.

Current technology permits three different strategies for enabling identities within the B4A system. These methodologies result in different costs and effort, and are not fully decentralized. The entities responsible for the system will need to select an appropriate technology depending on attributes of the to-be identified party.

- (A) For maintainers of the network, entities are preselected (by a trusted network partner, e.g., an NGO or local University) and introduced to the system [118]. With that, they will receive an on-chain identity which grants them the respective role in the network. As this process is expensive, only core maintainers of the network are onboarded with this approach.
- (B) For legal entities such as companies or research institutions which want to interact with the network, we recommend the party to create an identity which is linked to their domain. To enable a secure communication between users and websites, web servers usually have digital certificates, which allows the user to verify the correctness of the website⁶. We recommend an approach [120] that allows us to endorse on-chain identities with respective website certificates. This a) establishes a secure binding to a real-world identity and b) prevents the creation of multiple identities. This process can be further supervised by the maintainers of the network, e.g., that only research institutions can be registered providing a domain with a .edu Top Level Domain (TLD).

⁶These certificates are so-called TLS/SSL-certificates issued by Certificate Authorities. These certificate authorities ensure that only the website owner has access to this certificate. The existence of a certificate is shown by the "lock"-sign in the URL-bar in all modern browsers. These systems are used on over 90% of all websites according to Google [119]

TABLE IV Features of Blockchain

Key Blockchain Feature	Issues and Downsides
ν_1 Identity, attribution and non-repudiation	ι_1 Importance of attribution
	$\iota_{12.7}$ Lack of trust
ν_2 Provenance, tracking, and information verification	ι_3 Traceability of source of seeds and know how
	ι_{11} Lack of traceability of PGRs exchanged
	ι_7 Means of ensuring and documenting quality of seeds
	ι_8 Means ensuring quantity, availability and affordability
	ι_9 Knowledge creation and verification
ν_3 Disintermediation and cost reduction	$\iota_{12.6}$ High transaction costs

- (C) For farmers' collectives or for informal groups of farmers⁷ collectively making contributions to the system, it may be necessary to work through a trusted local NGO, local government body or local trusted University. In countries like India, for example, several recent movements aimed at reviving traditional ecological knowledge systems and farming techniques (such as Natural Farming) that help revive and improve agrobiodiversity, are pioneered by non-governmental organizations. These NGOs can then be the trusted third party through which legitimate data entries can be made into the B4A system. Novel systems such as Self-sovereign identities can be leveraged to issue such identities and onboard respective parties [121].
- (D) For private or natural persons (e.g., farmers and consumers), a KYC ("Know your Customer") process can be established. This is the most complex and costly approach, as the natural person who wants to interact with the system has to contact one of the maintainers of the network to get accredited for the network [122]. The maintainers need to ensure that the information about the natural person a) is correct and b) no duplicate entry exists. These maintainers are best placed near the location of the respective natural persons and comprise NGOs and other non-profit organisations.
- (E) If an entity (e.g., a farmer) has not the technical capabilities to participate in the network, other stakeholders need to act as a trustworthy intermediary between the entity and the network. With this approach the entity is entirely dependent on the intermediary, requiring that party to be trustworthy towards the original entity as well as towards the network.

With these processes in place, the aforementioned digital identity issues can be mitigated and digital identity can be securely used within the application.

2) ν_2 : Provenance, tracking and information verification: The append-only data structure of a blockchain creates a log of all activities in the network, such as trade between parties or the report of a new type of seed. A node within this network needs to evaluate all logs to get the current state of the network (e.g., what seeds are available). This allows the tracking and proof of provenance of data and values, as every action is transparent to the network. This also relates to the property of identity and non-repudiation, as a once created transaction that is logged in the network can never be removed from this log again. The main issue for tracking and provenance proofs in blockchain is the connection between on-chain data to their real-world counterpart, as properties of goods need to be measured, stored and potentially updated in the blockchain, requiring a) standardized methods of measurements and b) ways to ensure the trustworthiness of data respectively the entity that uploads the data [123]. This is often referred to as the Oracle problem [124].

The current research in blockchains focuses on either the establishment of digital twins for blockchain networks or secure oracles [124]–[126]. Trust in the data and the third party providing it can be achieved by:

- **Trust establishment as a form of consensus:** Multiple entities which are able to observe the same phenomenon are able to "vote" about the state of the phenomenon. This could be the case when multiple research institutes report the same types of quality for one specific seed. This consensus-driven approach can also work after-the-fact, as e.g, seed quality can be later on evaluated by multiple third parties which can then coordinate on the validity of previously submitted data [127].
- **Trust establishment due to tamper-proof hardware:** An alternative form of trust establishment requires the usage of certified hardware. This hardware is specifically designed for the use case and includes mechanisms to connect to the blockchain and submit collected data. For example, hardware that measures the temperature of shipped goods can be used to prove the cold chain. However, this approach leads to a centralization, as the hardware manufacturer controls the hardware and potentially can manipulate it, while also the hardware itself could be manipulated, depending on the complexity of the hardware (e.g., instead of cooling the seeds itself, only the sensor of the hardware could be cooled). Multiple hardware vendors can complicate attack vectors, providing for a more secure network [128].
- Establishment of a trusted third party: A trusted third party with the sole purpose of measuring real-world phenomena and providing this information on-chain is

⁷In all cases involving individual farmers or farmers collectives, the existence of banks well connected to the international banking network would be beneficial, if not crucial. Farmers who own bank accounts can then obtain direct payments into their accounts through the operation of smart contracts, on mutually agreed terms. This would reduce corruption and transaction costs.

also a viable way to go if parties exist within the network that are more trusted than others. A local research institute would be able to obtain reliable information about seeds, such that it can be directly put on the blockchain. Obviously, this also leads to some form of centralization, however the third party could still be blamed after the fact as all the submitted information is recorded [129].

This issue of trustworthy third party data is under research and even own networks like Chainlink [126] have been created to address aforementioned issues.

To resolve challenges within our use case, we recommend the adoption of a public permissioned blockchain and designing processes for coordination between multiple parties on the validity of data. In the proposed use case, multiple stakeholders exist which fulfill different roles in the system. As some parties might be more trustworthy than others (based on the role or on previous experience), a hierarchy can be established for determining the validity of data. For example, farmers should be able to claim properties of their seeds. However, other farmers as well as research institutions that have legitimately acquired a sample of the specific seeds should be able to verify or reject these properties, while also adding new, previously unknown features based on research (by research institutions or corporations) or observation (by farmers or consumers). Over time, an amazon like rating mechanism may have to be evolved with weights/ranking attributed to the rating based on track record - for example, the rating by a new end-consumer may be weighted/ranked lower than the rating by an established independent research institution. Proper management of roles and ratings can ensure validity of data and know-how entered into the system, or of materials transacted via the system.

3) ν_3 : Disintermediation and cost reduction: A blockchain as a decentralized network replaces centralized entities that charge fees for using their infrastructure. Removing third party intermediaries can, therefore, reduce costs [130]. However, this replacement comes with some caveats:

- The cost of removal is replaced with the cost to incentivize other parties to join and maintain the overall network, either by monetary incentives in public permissionless networks (e.g., Bitcoin) or through secondary incentives (being able to use the platform itself, caring for a social purpose or others) in permissioned networks. In the B4A use case, entities who run the network incur the costs of setting up and maintaining servers. Accordingly, these costs are not entirely removed but distributed among the participating parties. This could lead to reduced overall costs, especially in the long run when set up costs are recuperated.
- Not all activities of an intermediary can be replaced by a decentralized network. If the intermediary serves specific roles in the network e.g., as arbitrator, other trusted third parties have to be established or complex voting schemes have to be created to allow this feature in a decentralized network [131]. This is especially the case if we need to

decide about the data validity of given inputs. For that case, we need to establish means as mentioned in ν_2 .

- A true disintermediation is often only possible in use cases that permit or call for public permissionless networks. In permissioned networks such as B4A, some form of trusted intermediary is needed to establish, maintain and extend the quality and practical utility/value of the network over time. In some cases, such as in cases of farmers working via NGOs, the intermediary may also have control over the distribution of funds received in the network, as it makes the rules in the network and maintains it (for a specific group or regions, for example). In our case, the proposers and designers of this system need to involve all (local) stakeholders to find valid rules.
- The decentralization often comes with a limitation of throughput in the network. Public permissionless networks often face some technical barrier (e.g. 7 transactions per second in Bitcoin), while permissioned networks are only limited by the hardware and internet connection to other parties, as well as the complexity involved in verifying the quality of data/materials/know-how, at least to some extent, at the entry point [132].

For the B4A use case, the fair distribution of rights and responsibilities is important. We assume that due to the stakeholders of the network, no additional fees will occur e.g. on the level of farmers. They should be able to use the network without fees, as they provide key material to sustain and continuously add real value to the network.

VII. RQ3B - BUILDING INCENTIVES INTO B4A

A. Why Incentives?

A blockchain-based sustainable agrobiodiversity conservation system constitutes an institution for all involved stakeholders. Institutions provide the economic incentive structure that are expressed by informal constraints and formal rules [133]. Informal constraints such as sanctions or a certain code of conduct, as well we formal rules such as constitutions or law set the stage for how institutions and their incentive structures evolve. Thus, incentives provided in blockchainbased systems play an important role also for B4A and decide whether a system growth or stagnates. As we have seen, the traditional system was not able to align incentives, which led to a low agrodiversity conversation rate due to uncooperative incentive structures or even opportunistic behavior. Blockchain systems not only provide transparency, but also the certainty that agreed upon processes, payments, and actions will be executed. In other words, farmers can be certain that they gain the rewards that were agreed upon, which increases the trust in the system, and decreases the appearance of opportunistic behavior [134]. Thus, a blockchain-based B4A system allows for re-engineering not only the supply chain, but also incentive structure and payoff matrix for stakeholders such as farmers involved [135]. Therfore, it is plausible to assume that the existing inadequate intellectual property protection regimes that are not able to incentivize sustainable innovations can also be remodeled and rewards can be guaranteed once the system is blockchain-based.

Incentives can have different forms, ranging from assurance of compliance to user-functionalities or economic incentives such as personal benefits. Incentives can also nudge people to behave in a certain, intended way. In other words, blockchain systems can discourage people from actions which may adversely impact the sustainability of the overall system. As incentive mechanisms are part of the blockchain protocol, they reside in code on-chain to guarantee that given a certain behaviour is achieved, the agent, for example farmers and downstream researchers, can be certain that they receive the reward or any other benefit that has been promised. blockchain-based systems, supported by smart contracts and remote sensing algorithms, have been evaluated, for example, for reinforcing payments for ecosystem services in Namibia [136]. On-chain blockchain incentive mechanisms manifest incentives into blockchain systems using specific blockchain constructs, including game theoretical economic incentives, e.g., in form of validation rewards.

Blockchain systems are by nature distributed and have to align interests and incentives among the involved stakeholders. Alignment with the goals of the blockchain system occur when participants are incentivised to behave. In permissioned public blockchain systems, participants are pre-defined, which means that incentives can usually be created through traditional means (e.g., legal contracts, creating efficiencies and business revenue) [136].

B. Incentives necessary in B4A

In B4A, incentives are necessary for farmers to change their behavior toward a more biodiversity and soil sustaining farming. In blockchain systems supporting research and innovation with agrobiodiversity, a system-wide alignment of existing incentives or re-engineering of non-existing or contradictory incentives is required. In other words, every entity in the system should have as dominant strategy, to behave within the incentive structure engineered in the blockchain-based reorganized market. Incentive alignment has been achieved "when the system's embedded features induce users to employ the system consistent with the design objective" [137]. A blockchain-based agrobiodiversity system that aligns incentives across all entities involved allows agents to freely choose their own actions but uses incentives to make them inclined to choose actions that coincide with goals of the system's design. In such a blockchain system, a conscious and mindful incentive alignment is mission critical. Unless incentives are properly aligned, the nodes of the blockchain system will not contribute to form consensus.

Following the conclusions of [138], it appears that in B4A, blockchain can be used to

• incentivise farmers, researchers and other stakeholders to positively behave and interact in the system whilst simultaneously disincentivizing illicit behavior. This would, for example, incentivize proper declaration of source/origin and automate equitable benefit sharing with farmers;

- identifying (new) participants, making digital identity available and usable whilst ensuring that information once attributed cannot be repudiated, thereby reducing the chances of disputes resulting from dishonest behavior. This would, for example, ensure that any new user/stakeholder entering the system can be given a proper digital identity and all materials and know-how originating from her can be either be appropriately compensated/rewarded, or sent back/black listed (e.g., due to consistent low quality or false claims);
- maintaining an immutable record of transactions/transfers (e.g., record of agrobiodiversity and know-how transfers from farmers to various categories of end users⁸); this would, for example, ensure that in case of downstream research resulting in new products (e.g., improved, uniform seeds), royalties can be paid to farmer-originators.
- independently verify data entries to determine whether information submitted to the blockchain is correct in B4A. Instead of a proof of work based verification of transactions, it may be necessary to use other approaches such as "proof of authority".
- remove intermediaries in complex interaction processes and reduce overall transaction and communication costs. In B4A, while several regulatory intermediaries can be reduced, a trusted third party would still be necessary to check and confirm the quality and veracity of materials and know-how shared by various stakeholders. This entry gateway of corruption cannot be completely eliminated even when using blockchain. However, appropriate incentives written into smart contracts [136] can transform trusted third parties into stakeholders that can be rewarded for honest work.

VIII. CONCLUSIONS AND RECOMMENDATIONS FOR FUTURE RESEARCH

This paper identifies (i) concrete hurdles and loopholes in international regulations designed to protect and promote (in situ) agrobiodiversity conservation, (ii) practical hurdles faced by farmers engaged with farming using agrobiodiversity, and (iii) looks into into the extent to which blockchain technology may help in resolving the identified shortcomings and hurdles [19, 20].

Based on a review of the existing state of art, it appears that a blockchain based solutions can help overcome several of the hurdles in the current international regulations, and the resulting practical problems of (lack of) trust, traceability, and equitable data collection. For this, an appropriate software architecture needs to be implemented, including asymmetric cryptography, key-derivation algorithms and decentralised identity management. While high transaction costs currently

⁸A large number of stakeholders and end-users would potentially (need to) be involved in B4A, including farmers who are currently engaged in agriculture using agrobiodiversity; farmers who want to migrate to such agriculture; research institutes interested in conducting research or breeding with agrobiodiversity; corporations, including the seed industry; governmental regulators; seed and organic certification agencies and international regulatory authorities under the CBD and the Seed Treaty

prevent the equitable sharing of agrobiodiversity and associated know-how, blockchain solutions could lower costs by removing complex and expensive interaction processes. If complemented with biomarker technology, blockchain may allow equitable integration of digital sequence information in biodiversity regulations and thus facilitate implementation of more streamlined and transparent values and obligations for access and benefit sharing [89].

This feature, in fact, points to one of the crucial features of a blockchain based solution for the B4A usecase, namely, its ability to actually incentivize sharing of information, knowhow and materials. Since on-chain blockchain incentives mechanisms can guarantee rewards or other benefits for data or material shared, in a blockchain solution, incentives could empower data providers (e.g., farmers) to collectively track, control and monetize the usage of data and assets shared. The digital identity of stakeholders allows appropriate compensation and ensures the security and quality of information (maintaining an immutable record of transactions and verifying data entries). Challenges in relation to digital identity of stakeholders could be overcome, for example, by implementing pre-selected maintainers of the network, on-chain identities or "Know-yourcustomer" processes.

Blockchain based solutions, can, therefore, increase the number of legitimate transactios, minimize black-market or illegal transactions involving agrobiodiversity, and enhance incentives for research and innovation with agrobiodiversity conserved in situ. However, several current challenges cannot be addressed by blockchain (alone) and several issues may be aggravated by implementing a blockchain solution.

While in this paper we lay out the key features (in the form of challenges and shortcomings in existing systems) that have to inform such an application and it's ecosystem, as well as the possibilities and limitations of blockchain technology, a lot still needs to be done. This includes the definition of the software architecture, the exact roles and responsibilities of various "nodes" and stakeholders in the system, the concrete incentive mechanisms and the manner in which they can be implemented to keep all parties on board. Future work is, accordingly, aimed at close collaboration with all stakeholders to align interests with incentives and design concrete mechanisms to implement a blockchain for biodiversity.

In terms of real-world effects, technical solutions like blockchain for agrobiodiversity can help ensure that legal regimes and policies do not inequitably favor one direction of scientific research and innovation to the exclusion of others. Particularly, such solutions can incentivize research and innovation linked to long ignored Traditional Ecological Knowledge (TEK) systems that support sustainable in situ innovations with agrobiodivdrsity by both formal and informal sectors. Blockchain can, therefore, help diversify the directions of knowledge and value flows and create multiple options for successful farmer-migration to sustainable agriculture using local agrobiodiversity.

However, blockchain based solutions can also create new legal and ethical issues. Ethical issues linked to trustworthiness and integrity of codes and privacy concerns must be adequately addressed at an early stage of development. These issues are particularly significant with disruptive technologies like blockchain that can destabalize existing socio-economic power structures, including in rural and remote areas. Moving into a world ruled by code, one must also ensure that smart contracts that execute code are designed fairly, taking all stakeholders' interests equitably into account, i.e., they should be inclusive and free from bias. Empirical research may be necessary to identify what is considered 'fair and inclusive' by contributors (farming communities). With the emergence of "code" based governance, it would also be necessary to conduct extensive legal research to determine how issues of liability would be reconciled.

Further, in order to make the DLT or blockchain-based solutions more immediately usable in practical terms, smooth interaction between the developed system and existing governance structures and regulations is necessary. This would ensure a sustainable transition that secures meaningful and continuing interaction between human and autonomous actors. This goes also for the used currency in such a system. Instead of a cryptocurrency-based incentive mechanism, it may be necessary to adopt a "Biodiversity points" based reward system - similar to systems adopted in carbon trading. In countries like India, where a biodiversity fund exists under the Biodiversity Act, these points could then be exchanged for cash by farmers and other stakeholders. The details of such a system, however, require more research in the future.

In a blockchain for biodiversity world, all stakeholders, farming systems and sectors could have a fair(er) chance of surviving and flourishing equitably. Most importantly, all stakeholders could be incentivized to partake in research and sustainable innovations that enhance agrobiodiversity in situ. Concerted, multi-disciplinary and translational research is needed to accomplish this.

REFERENCES

- FAO, "What is agrobiodiversity?" 1999. [Online]. Available: http: //www.fao.org/3/y5609e/y5609e01.htm
- [2] L. E. Jackson, U. Pascual, and T. Hodgkin, "Utilizing and conserving agrobiodiversity in agricultural landscapes," *Agriculture, ecosystems environment*, vol. 121, no. 3, pp. 196–210, 2007.
- [3] M. G. Van Der Heijden, S. De Bruin, L. Luckerhoff, R. S. Van Logtestijn, and K. Schlaeppi, "A widespread plant-fungal-bacterial symbiosis promotes plant biodiversity, plant nutrition and seedling recruitment," *The ISME journal*, vol. 10, no. 2, p. 389, 2016.
- [4] A. Fatima, M. Farid, K. Safdar, A. Fayyaz, S. M. Ali, S. Adnan, M. Nawaz, H. Munir, N. Raza, and M. Zubair, *Loss of Agro-Biodiversity and Productivity Due to Climate Change in Continent Asia: A Review.* Springer, 2020, pp. 51–71.
- [5] J. Witcombe, P. Hollington, C. Howarth, S. Reader, and K. Steele, "Breeding for abiotic stresses for sustainable agriculture," *Philosophi*cal Transactions of the Royal Society B: Biological Sciences, vol. 363, no. 1492, pp. 703–716, 2008.
- [6] 1992. [Online]. Available: https://www.cbd.int/doc/legal/cbd-en.pdf
- [7] FAO, "International treaty on plant genetic resources for food and agriculture," 2009. [Online]. Available: http://www.fao.org/3/i0510e/ I0510E.pdf
- [8] F. Fusi, E. W. Welch, and M. Siciliano, "Barriers and facilitators of access to biological material for international research: The role of institutions and networks," *Science and Public Policy*, vol. 46, no. 2, pp. 275–289, 2019.

- [9] S. Jinnah and S. Jungcurt, "Could access requirements stifle your research," *Science*, vol. 323, no. 5913, pp. 464–465, 2009.
- [10] FAO, "What is happening to agrobiodiversity?" 1999. [Online]. Available: http://www.fao.org/3/y5609e/y5609e02.htm
- [11] J. E. Pérez-Jaramillo, V. J. Carrión, M. de Hollander, and J. M. Raaijmakers, "The wild side of plant microbiomes," *Microbiome*, vol. 6, no. 1, p. 143, 2018.
- [12] M. A. Tsiafouli, E. Thébault, S. P. Sgardelis, P. C. De Ruiter, W. H. Van Der Putten, K. Birkhofer, L. Hemerik, F. T. De Vries, R. D. Bardgett, and M. V. Brady, "Intensive agriculture reduces soil biodiversity across europe," *Global change biology*, vol. 21, no. 2, pp. 973–985, 2015.
- [13] A. Orgiazzi, R. D. Bardgett, and E. Barrios, *Global soil biodiversity atlas*. European Commission, 2016.
- [14] E. D. Vià, "Seed diversity, farmers' rights, and the politics of repeasantization," *International Journal of Sociology of Agriculture Food*, vol. 19, no. 2, 2012.
- [15] T. Wattnem, "Seed laws, certification and standardization: outlawing informal seed systems in the global south," *The Journal of Peasant Studies*, vol. 43, no. 4, pp. 850–867, 2016.
- [16] K. Peschard and S. Randeria, "'keeping seeds in our hands': the rise of seed activism," *The Journal of Peasant Studies*, pp. 1–33, 2020.
- [17] European Commission.
- [18] E. Commission, "Organic farming in the eu," 2019. [Online]. Available: https://ec.europa.eu/info/sites/info/files/food-farming-fisheries/ farming/documents/market-brief-organic-farming-in-the-eu_mar2019_ en.pdf
- [19] M. Kochupillai, Promoting sustainable innovations in plant varieties. Springer, 2016, vol. 5.
- [20] —, Is UPOV 1991 a good fit for developing countries? Edward Elgar, 2019a.
- [21] M. Kochupillai, G. Radick, P. Rao, N. Kopytko, J. Köninger, and J. Matthiessen, "Sustainable seed innovation - white paper for the indian government," 2019.
- [22] H. Zhang, N. Mittal, L. J. Leamy, O. Barazani, and B. Song, "Back into the wild—apply untapped genetic diversity of wild relatives for crop improvement," *Evolutionary applications*, vol. 10, no. 1, pp. 5–24, 2017.
- [23] H. Dempewolf, G. Baute, J. Anderson, B. Kilian, C. Smith, and L. Guarino, "Past and future use of wild relatives in crop breeding," *Crop Science*, vol. 57, no. 3, pp. 1070–1082, 2017.
- [24] F. Batur and T. Dedeurwaerdere, "The use of agrobiodiversity for plant improvement and the intellectual property paradigm: institutional fit and legal tools for mass selection, conventional and molecular plant breeding," *Life sciences, society and policy*, vol. 10, no. 1, pp. 1–29, 2014.
- [25] N. Kopytko, "Supporting sustainable innovations: An examination of india farmer agro-biodiversity conservation," *Journal of Environment* and Development, vol. 28(4), pp. 386–411, 2019.
- [26] O. T. Coomes, S. J. McGuire, E. Garine, S. Caillon, D. McKey, E. Demeulenaere, D. Jarvis, G. Aistara, A. Barnaud, and P. Clouvel, "Farmer seed networks make a limited contribution to agriculture? four common misconceptions," *Food Policy*, vol. 56, pp. 41–50, 2015.
- [27] R. Salazar, N. P. Louwaars, and B. Visser, "Protecting farmers' new varieties: new approaches to rights on collective innovations in plant genetic resources," *World Development*, vol. 35, no. 9, pp. 1515–1528, 2007.
- [28] F. Abay, A. Waters-Bayer, and Bjørnstad, "Farmers' seed management and innovation in varietal selection: implications for barley breeding in tigray, northern ethiopia," *Ambio*, pp. 312–320, 2008.
- [29] S. McGuire and L. Sperling, "Seed systems smallholder farmers use," *Food Security*, vol. 8, no. 1, pp. 179–195, 2016.
- [30] A. Nyong, F. Adesina, and B. O. Elasha, "The value of indigenous knowledge in climate change mitigation and adaptation strategies in the african sahel," *Mitigation and Adaptation strategies for global Change*, vol. 12, no. 5, pp. 787–797, 2007.
- [31] E. C. Nkomwa, M. K. Joshua, C. Ngongondo, M. Monjerezi, and F. Chipungu, "Assessing indigenous knowledge systems and climate change adaptation strategies in agriculture: A case study of chagaka village, chikhwawa, southern malawi," *Physics and Chemistry of the Earth, Parts A/B/C*, vol. 67, pp. 164–172, 2014.
- [32] P. Mapfumo, F. Mtambanengwe, and R. Chikowo, "Building on indigenous knowledge to strengthen the capacity of smallholder farming communities to adapt to climate change and variability in southern africa," *Climate and Development*, vol. 8, no. 1, pp. 72–82, 2016.

- [33] M. Halewood, T. Chiurugwi, R. Sackville Hamilton, B. Kurtz, E. Marden, E. Welch, F. Michiels, J. Mozafari, M. Sabran, and N. Patron, "Plant genetic resources for food and agriculture: opportunities and challenges emerging from the science and information technology revolution," *New Phytologist*, vol. 217, no. 4, pp. 1407–1419, 2018.
- [34] S. B. Brush, "Farmers' rights and protection of traditional agricultural knowledge," *World Development*, vol. 35, no. 9, pp. 1499–1514, 2007.
- [35] R. G. Kochupillai, Mrinalini, "А wake-up call on proprietary seeds," The Hindu, 2019. [Online]. Available: https://www.thehindu.com/opinion/op-ed/ a-wake-up-call-on-proprietary-seeds/article27072442.ece
- [36] E. O. Wilson, "Threats to biodiversity," *Scientific American*, vol. 261, no. 3, pp. 108–117, 1989.
- [37] K. G. Rosendal, I. Olesen, H. B. Bentsen, M. W. Tvedt, and M. Bryde, "Access to and legal protection of aquaculture genetic resources—norwegian perspectives," *The Journal of World Intellectual Property*, vol. 9, no. 4, pp. 392–412, 2006.
- [38] E. Morgera, "The need for an international legal concept of fair and equitable benefit sharing," *European Journal of International Law*, vol. 27, no. 2, pp. 353–383, 2016.
- [39] 2012. [Online]. Available: https://www.cbd.int/intro/
- [40] 1992. [Online]. Available: https://www.cbd.int/doc/legal/cbd-en.pdf
- [41] I. Sundh, A. Wilcks, and M. S. Goettel, Beneficial microorganisms in agriculture, food and the environment: safety assessment and regulation. CABI, 2012.
- [42] Bjørnstad, S. Tekle, and M. Göransson, ""facilitated access" to plant genetic resources: does it work?" *Genetic resources and crop evolution*, vol. 60, no. 7, pp. 1959–1965, 2013.
- [43] J. H. Reichman, "Why the nagoya protocol to the cbd matters to science and industry in canada and the united states," 2018.
- [44] K. D. Prathapan, R. Pethiyagoda, K. S. Bawa, P. H. Raven, and P. D. Rajan, "When the cure kills—cbd limits biodiversity research," *Science*, vol. 360, no. 6396, pp. 1405–1406, 2018.
- [45] CBD, "Nagoya protocol on access to genetic resources and the fair and equitable sharing of benefits arising from their utilization to the convention on biological diversity," 2014. [Online]. Available: https://www.cbd.int/abs/doc/protocol/nagoya-protocol-en.pdf
- [46] F. Nations and A. O. of the United, "International treaty on plant genetic resources for food and agriculture," 2004. [Online]. Available: http://www.fao.org/3/i0510e/I0510E.pdf
- [47] N. Pauchard, "Access and benefit sharing under the convention on biological diversity and its protocol: What can some numbers tell us about the effectiveness of the regulatory regime?" *Resources*, vol. 6, no. 1, p. 11, 2017.
- [48] E. C. Kamau, G. Winter, and P.-T. Stoll, Research and development on genetic resources: public domain approaches in implementing the nagoya protocol. Routledge, 2015.
- [49] 2020. [Online]. Available: http://www.fao.org/plant-treaty/ areas-of-work/the-multilateral-system/overview/en/
- [50] Bjørnstad, S. Tekle, and M. Göransson, ""facilitated access" to plant genetic resources: does it work?" *Genetic resources and crop evolution*, vol. 60, no. 7, pp. 1959–1965, 2013.
- [51] D. Schaffrin, B. Görlach, C. Gerstetter, S. Jungcurt, and M. Collins, "The international treaty on plant genetic resources for food and agriculture-implications for developing countries and interdependence with international biodiversity and intellectual property law," *Final Report IPDEV Work Package*, vol. 5, pp. 8–15, 2006.
- [52] E. Kamau, "The multilateral of the international treaty on plant genetic resources for food and agriculture: Lessons and room for further development," *Common Pools of Genetic Resources. Equity and Innovation in International Biodiversity Law, Earthscan-Routledge, London*, pp. 343–372, 2013.
- [53] J. Esquinas-Alcázar, "Protecting crop genetic diversity for food security: political, ethical and technical challenges," *Nature Reviews Genetics*, vol. 6, no. 12, p. 946, 2005.
- [54] S. Laird, R. Wynberg, M. Rourke, F. Humphries, M. R. Muller, and C. Lawson, "Rethink the expansion of access and benefit sharing," *Science*, vol. 367, no. 6483, pp. 1200–1202, 2020.
- [55] B. De Jonge, "What is fair and equitable benefit-sharing?" Journal of Agricultural Environmental Ethics, vol. 24, no. 2, pp. 127–146, 2011. [Online]. Available: <GotoISI>://WOS:000287458900004
- [56] F. Jankowski, S. Louafi, N. A. Kane, M. Diol, A. D. Camara, J. L. Pham, C. Berthouly-Salazar, and A. Barnaud, "From texts to enacting practices: defining fair and equitable research principles for plant

genetic resources in west africa," Agriculture and Human Values, 2020. [Online]. Available: <GotoISI>://WOS:000531229500001

- [57] J. Esquinas-Alcázar, "Protecting crop genetic diversity for food security: political, ethical and technical challenges," *Nature Reviews Genetics*, vol. 6, no. 12, pp. 946–953, 2005.
- [58] J. Martins, D. Cruz, and V. Vasconcelos, "The nagoya protocol and its implications on the eu atlantic area countries," *Journal of Marine Science and Engineering*, vol. 8, no. 2, pp. 92–92, 2020. [Online]. Available: <GotoISI>://CCC:000519244200081
- [59] M. Heinrich, F. Scotti, A. Andrade-Cetto, M. Berger-Gonzalez, J. Echeverria, F. Friso, F. Garcia-Cardona, A. Hesketh, M. Hitziger, C. Maake, M. Politi, C. Spadafora, and R. Spadafora, "Access and benefit sharing under the nagoya protocol-quo vadis? six latin american case studies assessing opportunities and risk," *Frontiers in Pharmacology*, vol. 11, 2020. [Online]. Available: <GotoISI>://WOS:000543514900001
- [60] G. Aviles-Polanco, D. J. Jefferson, M. A. Almendarez-Hernandez, and L. F. Beltran-Morales, "Factors that explain the utilization of the nagoya protocol framework for access and benefit sharing," *Sustainability*, vol. 11, no. 20, pp. 5550–5550, 2019. [Online]. Available: <GotoISI>://CCC:000498398900007
- [61] P. Oldham, S. Hall, and O. Forero, "Biological diversity in the patent system," *Plos One*, vol. 8, no. 11, 2013. [Online]. Available: <GotoISI>://WOS:000327252100047
- [62] A. Sirakaya, "Balanced options for access and benefit-sharing: Stakeholder insights on provider country legislation," *Frontiers in Plant Science*, vol. 10, pp. 1175–1175, 2019. [Online]. Available: <GotoISI>://CCC:000497561100001
- [63] —, "Balanced options for access and benefit-sharing: Stakeholder insights on provider country legislation," *Frontiers in Plant Science*, vol. 10, 2019. [Online]. Available: <GotoISI>://WOS:000497561100001
- [64] L. Silvestri, A. Sosa, F. Mc Kay, M. D. Vitorino, M. Hill, C. Zachariades, S. Hight, P. Weyl, D. Smith, D. Djeddour, and P. G. Mason, "Implementation of access and benefit-sharing measures has consequences for classical biological control of weeds," *Biocontrol*, vol. 65, no. 2, pp. 125–141, 2020. [Online]. Available: <GotoISI>://CCC:000520211100001
- [65] D. Schroeder, R. Chennells, C. Louw, L. Snyders, and T. Hodges, "The rooibos benefit sharing agreement-breaking new ground with respect, honesty, fairness, and care," *Cambridge Quarterly of Healthcare Ethics*, vol. 29, no. 2, pp. 285–301, 2020. [Online]. Available: <GotoISI>://WOS:000524929100012
- [66] E. Tsioumani, "Itpgrfa gb-8: A missed opportunity for multilateralism," *Environmental Policy and Law*, vol. 49, no. 6, pp. 320–323, 2019.
- [67] W. Talaat, Protection of the Associated Traditional Knowledge on Genetic Resources: Beyond the Nagoya Protocol, ser. Procedia Social and Behavioral Sciences, 2013, vol. 91, pp. 673–678. [Online]. Available: <GotoISI>://WOS:000348252800074
- [68] M. C. C. Segger and F. K. Phillips, "Indigenous traditional knowledge for sustainable development: the biodiversity convention and plant treaty regimes," *Journal of Forest Research*, vol. 20, no. 5, pp. 430–437, 2015. [Online]. Available: <GotoISI>://WOS:000360863200003
- [69] T. Ulian, M. Diazgranados, S. Pironon, S. Padulosi, U. Liu, L. Davies, M. R. Howes, J. S. Borrell, I. Ondo, and O. A. Pérez-Escobar, "Unlocking plant resources to support food security and promote sustainable agriculture," *Plants, People, Planet*, vol. 2, no. 5, pp. 421– 445, 2020.
- [70] F. Rabitz, "Access without benefit-sharing: design, effectiveness and reform of the fao seed treaty," *International Journal of the Commons*, vol. 11, no. 2, pp. 621–640, 2017. [Online]. Available: <GotoISI>://CCC:000413186700002
- [71] C. Lawson, M. Rourke, and F. Humphries, "Information as the latest site of conflict in the ongoing contests about access to and sharing the benefits from exploiting genetic resources," *Queen Mary Journal* of Intellectual Property, vol. 10, no. 1, pp. 7–33, 2020. [Online]. Available: <GotoISI>://CCC:000560613000002
- [72] F. Rabitz, "Biopiracy after the nagoya protocol: problem structure, regime design and implementation challenges," *Brazilian Political Science Review*, vol. 9, no. 2, pp. 30–53, 2015.
- [73] D. Neumann, A. Borisenko, J. Coddington, C. Häuser, C. Butler, A. Casino, J. Vogel, G. Haszprunar, and P. Giere, "Global biodiversity research tied up by juridical interpretations of access and benefit

sharing," Organisms Diversity Evolution, vol. 18, no. 1, pp. 1-12, 2018.

- [74] E. W. Welch, E. Shin, and J. Long, "Potential effects of the nagoya protocol on the exchange of non-plant genetic resources for scientific research: Actors, paths, and consequences," *Ecological Economics*, vol. 86, pp. 136–147, 2013.
- [75] K. Davis, M. F. Smit, M. Kidd, S. Sharrock, and P. Allenstein, "An access and benefit-sharing awareness survey for botanic gardens: Are they prepared for the nagoya protocol?" *South African Journal of Botany*, vol. 98, pp. 148–156, 2015.
- [76] E. W. Welch, F. Fusi, S. Louafi, and M. Siciliano, "Genetic resource policies in international collaborative research for food and agriculture: A study of usaid-funded innovation labs," *Global food security*, vol. 15, pp. 33–42, 2017.
- [77] 2020. [Online]. Available: https://seedworld.com/ international-treaty-on-plant-genetic-resources-smolders/,2020
- [78] D. Neumann, A. Borisenko, J. Coddington, C. Häuser, C. Butler, A. Casino, J. Vogel, G. Haszprunar, and P. Giere, "Global biodiversity research tied up by juridical interpretations of access and benefit sharing," *Organisms Diversity & Evolution*, vol. 18, no. 1, pp. 1–12, 2018.
- [79] M. J. Oliva, "Sharing the benefits of biodiversity: A new international protocol and its implications for research and development," *Planta Medica*, vol. 77, no. 11, pp. 1221–1227, 2011. [Online]. Available: <GotoISI>://CCC:000293007700017
- [80] T. Dedeurwaerdere, P. Melindi-Ghidi, and A. Broggiato, "Global scientific research commons under the nagoya protocol: Towards a collaborative economy model for the sharing of basic research assets," *Environmental Science Policy*, vol. 55, pp. 1–10, 2016. [Online]. Available: <GotoISI>://CCC:000366775200001
- [81] A. Broggiato, "Abs and food security," *Environmental Policy and Law*, vol. 39, no. 6, p. 287, 2009.
- [82] M. Chevarria-Lazo, D. Bazile, D. Dessauw, S. Louafi, M. Trommetter, and H. Hocdé, "Quinoa and the exchange of genetic resources: Improving the regulation systems," 2015.
- [83] 2020. [Online]. Available: https://www.cbd.int/doc/c/fef9/2f90/ 70f037ccc5da885dfb293e88/dsi-ahteg-2020-01-03-en.pdf
- [84] C. Lawson and M. Rourke, "Digital sequence information as a marine genetic resource under the proposed unclos legally binding instrument," *Marine Policy*, p. 103878, 2020.
- [85] H. M. Haugen, "The un declaration on peasants' rights (undrop): Is article 19 on seed rights adequately balancing intellectual property rights and the right to food?" *The Journal of World Intellectual Property*, 2020.
- [86] S. Aubry, "The future of digital sequence information for plant genetic resources for food and agriculture," *Frontiers in plant science*, vol. 10, p. 1046, 2019.
- [87] ITPGRFA, "Seventh session of the governing body of the itpgrfa," 2017. [Online]. Available: http://www.fao.org/3/MV606/mv606.pdf.
- [88] —, "Submission of views and information on digital sequence informationsecretariat of theinternational treaty on plant genetic resources for food and agriculture," 2019. [Online]. Available: https://www.cbd.int/abs/DSI-views/2019/ITPGRFA-DSI.pdf
- [89] J. Ambler, A. A. Diallo, P. K. Dearden, P. Wilcox, M. Hudson, and N. Tiffin, "Including digital sequence data in the nagoya protocol can promote data sharing," *Trends in biotechnology*, 2020.
- [90] Bjørnstad, "'do not privatize the giant's shoulders': rethinking patents in plant breeding," *Trends in biotechnology*, vol. 34, no. 8, pp. 609–617, 2016.
- [91] S. Nakamoto, "Bitcoin: A peer-to-peer electronic cash system," 2008.
- [92] M. Crosby, P. Pattanayak, S. Verma, and V. Kalyanaraman, "Blockchain technology: Beyond bitcoin," *Applied Innovation*, vol. 2, no. 6-10, p. 71, 2016.
- [93] R. G. Brown, "R3 Corda: What makes it different | Richard Gendal Brown." [Online]. Available: https://gendal.me/2016/10/25/ r3-corda-what-makes-it-different/
- [94] M. Swan, Blockchain: Blueprint for a new economy. "O'Reilly Media, Inc.", 2015.
- [95] Z. Zheng, S. Xie, H.-N. Dai, X. Chen, and H. Wang, "Blockchain challenges and opportunities: A survey," *International Journal of Web* and Grid Services, vol. 14, no. 4, pp. 352–375, 2018.
- [96] U. Gallersdörfer and F. Matthes, "Towards valid use cases: Requirements and supporting characteristics of proper blockchain applica-

tions," in 2020 Seventh International Conference on Software Defined Systems (SDS). IEEE, 2020, pp. 202–207.

- [97] K. Wüst and A. Gervais, "Do you need a blockchain?" in 2018 Crypto Valley Conference on Blockchain Technology (CVCBT). IEEE, 2018, pp. 45–54.
- [98] E. M. Abou-Nassar, A. M. Iliyasu, P. M. El-Kafrawy, O. Song, A. K. Bashir, and A. A. A. El-Latif, "Ditrust chain: Towards blockchainbased trust models for sustainable healthcare iot systems," *IEEE Access*, vol. 8, pp. 111 223–111 238, 2020.
- [99] C. Stoll, L. Klaaßen, and U. Gallersdörfer, "The Carbon Footprint of Bitcoin," *Joule*, vol. 3, no. 7, 2019.
- [100] U. Gallersdörfer, L. Klaaßen, and C. Stoll, "Energy consumption of cryptocurrencies beyond bitcoin," *Joule*, vol. 4, no. 9, pp. 1843–1846, 2020.
- [101] coinmarketcap.com, "CryptoCurrency Market Capitalizations," \url{https://coinmarketcap.com/}, 2017.
- [102] A. Tewari and B. Gupta, "Security, privacy and trust of different layers in internet-of-things (iots) framework," *Future Generation Computer Systems*, vol. 108, pp. 909–920, 2020. [Online]. Available: https: //www.sciencedirect.com/science/article/pii/S0167739X17321003
- [103] A. Kamilaris, A. Fonts, and F. X. Prenafeta-Bold\$\acute\nu\$, "The rise of blockchain technology in agriculture and food supply chains," *Trends* in Food Science & Technology, vol. 91, pp. 640–652, 2019.
- [104] U. Gallersdörfer and F. Matthes, "Tamper-proof volume tracking in supply chains with smart contracts," in *European Conference on Parallel Processing*. Springer, 2018, pp. 367–378.
- [105] S. S. Kamble, A. Gunasekaran, and R. Sharma, "Modeling the blockchain enabled traceability in agriculture supply chain," *International Journal of Information Management*, vol. 52, p. 101967, 2020.
- [106] P. De Bernardi and D. Azucar, "Innovative and Sustainable Food Business Models," in *Innovation in Food Ecosystems*. Springer, 2020, pp. 189–221.
- [107] V. S. Yadav and A. R. Singh, "A Systematic Literature Review of Blockchain Technology in Agriculture," in *Proceedings of the International Conference on Industrial Engineering and Operations Management*, 2019, pp. 973–981.
- [108] D. Davcev, L. Kocarev, A. Carbone, V. Stankovski, and K. Mitresk, "Blockchain-based Distributed Cloud/Fog Platform for IoT Supply Chain Management," in *Eighth international conference on advances* in computing, electronics and electrical technology (CEET), 2018, pp. 51–58.
- [109] B. Gupta and M. Quamara, "An overview of internet of things (iot): Architectural aspects, challenges, and protocols," *Concurrency and Computation: Practice and Experience*, vol. 32, no. 21, p. e4946, 2020.
- [110] A. P. Plageras, K. E. Psannis, C. Stergiou, H. Wang, and B. B. Gupta, "Efficient iot-based sensor big data collection–processing and analysis in smart buildings," *Future Generation Computer Systems*, vol. 82, pp. 349–357, 2018.
- [111] J. Abadi and M. Brunnermeier, "Blockchain economics," National Bureau of Economic Research, Tech. Rep., 2018.
- [112] B. Notheisen, J. B. Cholewa, and A. P. Shanmugam, "Trading Real-World Assets on Blockchain," *Business & Information Systems Engineering*, vol. 59, no. 6, pp. 425–440, 2017. [Online]. Available: https://doi.org/10.1007/s12599-017-0499-8
- [113] V. L. Lemieux, "Blockchain and distributed ledgers as trusted recordkeeping systems," in *Future Technologies Conference (FTC)*, vol. 2017, 2017.
- [114] R. Kamath, "Food traceability on blockchain: Walmart's pork and mango pilots with IBM," *The Journal of the British Blockchain Association*, vol. 1, no. 1, p. 3712, 2018.
- [115] A. B. Pedersen, M. Risius, and R. Beck, "A ten-step decision path to determinewhen to use blockchain technologies," *MIS Quarterly Executive*, vol. 18, no. 2, p. 3, 2019.
- [116] S. Y. Lim, P. T. Fotsing, A. Almasri, O. Musa, M. L. M. Kiah, T. F. Ang, and R. Ismail, "Blockchain technology the identity management and authentication service disruptor: a survey," *Int. J. Adv. Sci. Eng. Inf. Technol*, vol. 8, no. 4-2, pp. 1735–1745, 2018.
- [117] R. L. Rivest, A. Shamir, and L. Adleman, "A method for obtaining digital signatures and public-key cryptosystems," *Communications of the ACM*, vol. 21, no. 2, pp. 120–126, 1978.
- [118] R. Viswanathan, D. Dasgupta, and S. Govindaswamy, "Blockchain solution reference architecture (bsra)," *IBM Journal of Research and Development*, vol. 63, no. 2/3, pp. 1–1, 2019.

- [119] "HTTPS-Verschlüsselung im Web Google Transparenzbericht." [Online]. Available: https://transparencyreport.google.com/https/overview
- [120] U. Gallersdörfer and F. Matthes, "AuthSC: Mind the Gap between Web and Smart Contracts," 4 2020. [Online]. Available: http: //arxiv.org/abs/2004.14033
- [121] R. Soltani, U. T. Nguyen, and A. An, "A new approach to client onboarding using self-sovereign identity and distributed ledger," in 2018 IEEE International Conference on Internet of Things (iThings) and IEEE Green Computing and Communications (GreenCom) and IEEE Cyber, Physical and Social Computing (CPSCom) and IEEE Smart Data (SmartData). IEEE, 2018, pp. 1129–1136.
- [122] J. P. Moyano and O. Ross, "Kyc optimization using distributed ledger technology," *Business & Information Systems Engineering*, vol. 59, no. 6, pp. 411–423, 2017.
- [123] B. Notheisen, J. B. Cholewa, and A. P. Shanmugam, "Trading realworld assets on blockchain," *Business & Information Systems Engineering*, vol. 59, no. 6, pp. 425–440, 2017.
- [124] A. Egberts, "The oracle problem-an analysis of how blockchain oracles undermine the advantages of decentralized ledger systems," *Available* at SSRN 3382343, 2017.
- [125] S. Huang, G. Wang, Y. Yan, and X. Fang, "Blockchain-based data management for digital twin of product," *Journal of Manufacturing Systems*, vol. 54, pp. 361–371, 2020.
- [126] S. Ellis, A. Juels, and S. Nazarov, "Chainlink a decentralized oracle network," *Retrieved March*, vol. 11, p. 2018, 2017.
- [127] S. Bano, A. Sonnino, M. Al-Bassam, S. Azouvi, P. McCorry, S. Meiklejohn, and G. Danezis, "Sok: Consensus in the age of blockchains," in *Proceedings of the 1st ACM Conference on Advances in Financial Technologies*, 2019, pp. 183–198.
- [128] A. Schaad, T. Reski, and O. Winzenried, "Integration of a secure physical element as a trusted oracle in a hyperledger blockchain." in *ICETE* (2), 2019, pp. 498–503.
- [129] X. Xu, I. Weber, and M. Staples, Architecture for blockchain applications. Springer, 2019.
- [130] L. García-Bañuelos, A. Ponomarev, M. Dumas, and I. Weber, "Optimized execution of business processes on blockchain," in *International Conference on Business Process Management*. Springer, 2017, pp. 130–146.
- [131] B. Waltl, C. Sillaber, U. Gallersdörfer, and F. Matthes, "Blockchains and smart contracts: a threat for the legal industry?" in *Business Transformation through Blockchain*, Springer, 2019, pp. 287–315.
- [132] F. Velde et al., "Bitcoin: A primer," 2013.
- [133] D. North, "Institutions stör," *The Journal of Economic Perspectives*, vol. 5, no. 1, pp. 97–112, 1991.
- [134] Y. He, H. Li, X. Cheng, Y. Liu, C. Yang, and L. Sun, "A blockchain based truthful incentive mechanism for distributed p2p applications," *IEEE Access*, vol. 6, pp. 27324–27335, 2018.
- [135] Z. Liu, N. C. Luong, W. Wang, D. Niyato, P. Wang, Y.-C. Liang, and D. I. Kim, "A survey on applications of game theory in blockchain," *arXiv preprint arXiv:1902.10865*, 2019.
- [136] D. Oberhauser, "Blockhain for environmental governance: Can smart contracts reinforce payments for ecosystem services in namibia?" *Front. Blockchain*, vol. 2, 2020.
- [137] S. Ba, J. Stallaert, and A. B. Whinston, "Research commentary: introducing a third dimension in information systems design—the case for incentive alignment," *Information systems research*, vol. 12, no. 3, pp. 225–239, 2001.
- [138] L. Hughes, Y. K. Dwivedi, S. K. Misra, N. P. Rana, V. Raghavan, and V. Akella, "Blockchain research, practice and policy: Applications, benefits, limitations, emerging research themes and research agenda," *International Journal of Information Management*, vol. 49, pp. 114–129, 2019. [Online]. Available: http://www.sciencedirect.com/ science/article/pii/S0268401219302014